Author's response to reviews

Title: Comparative cost-efficiency of EVOTECH ECR versus Medivator DSD-201 in a real-world Canadian hospital endoscopy setting.

Authors:

Lindy Forte (lindy.forte@sympatico.ca)
Cynthia Shum (cshum@its.jnj.com)

Version: 2 Date: 4 January 2011

Author's response to reviews:

Dear Editors,

Additional information has been added to the Competing Interests Section to describe the relationship between the first author and VALORE Research.

Kind regards,

Lindy Forte

VALORE Research

Toronto, Canada

416-453-8350